Author's response to reviews

Title: The effectiveness of a trauma-focused psycho-educational secondary prevention program for children exposed to interparental violence: study protocol for a randomized controlled trial

Authors:

Mathilde M. Overbeek (m.m.overbeek@vu.nl)
J. Clasien de Schipper (j.c.de.schipper@vu.nl)
Francien Lamers-Winkelman (f.lamers@vu.nl)
Carlo Schuengel (c.schuengel@vu.nl)

Version: 3 Date: 24 January 2012

Author's response to reviews: see over
Dear Sir/Madam,

We are delighted you have accepted our manuscript for publication in BMC Trials. As requested we made a few formatting corrections:

- The title of the manuscript has been changed from ‘The effectiveness of a psycho-educational secondary prevention program for children exposed to interparental violence: An RCT evaluating the need for trauma-focused interventions’ to ‘The effectiveness of a trauma-focused psycho-educational secondary prevention program for children exposed to interparental violence: study protocol for a randomized controlled trial.’
- The abstract has been shortened from 352 to 350 words.
- Three typographic errors have been corrected (p. 11 enrol -> enroll, p. 12 aid -> assist, p. 34 Oppenheid -> Oppenheim).
- At p. 31 the English translation of the control program has been added.
- The acknowledgement of L.M. van den Berg has been changed to: ‘... and L. M. Van den Berg, MSc1,2 for her help with the implementation of the control program and data collection.

We hope we have informed you correctly with this overview. If you have any questions concerning the study or manuscript, please do not hesitate to contact me by phone or email.

Sincerely,

Mathilde M. Overbeek, PhD student
VU University, Amsterdam
Department of Clinical Child and Family Studies
Van der Boechorststraat 1
1081 BT Amsterdam
phone: 0031 (0)20 598 8891
m.m.overbeek@vu.nl